Author's response to reviews

Title: Triple-negative breast cancer is associated with EGFR, CK5/6 and c-KIT expression in Malaysian women

Authors:

Shant Kishen Kanapathy Pillai (shantkishen@gmail.com)
Annie Tay (annie_tay@imu.edu.my)
Suseela Nair (suseelanr@gmail.com)
Chee-Onn Leong (cheeonn_leong@imu.edu.my)

Version: 2 Date: 2 February 2012

Author's response to reviews: see over
Author’s response to editors comments of the paper

“Triple-negative breast cancer is associated with EGFR, CK5/6 and c-KIT expression in Malaysian women”

(Manuscript: MS 7183550976492164)

Shant Kishen Kanapathy Pillai, Annie Tay, Suseela Nair, Chee-Onn Leong

Editors’ comment:

Please confirm that the patients gave written consent for the samples to be included in the study, and add a statement to that effect in the Methods section of the manuscript.

Authors’ response:

The authors wish to thank the editors for their time and thoughtful comments. Written informed consent for use of all human specimens in this study was obtained at the time of enrollment. This statement is now added to the Methods section of the revised manuscript.

Editors’ comment:

Please include an Authors’ Contributions section before the Acknowledgements and Reference list.

Authors’ response:

The Authors’ Contributions section is now added in the revised manuscript.
We hope that the editors find that his/her concerns have been adequately addressed in the revised article, and in this response.

Sincerely,

Chee-Onn Leong
Associate Professor (Ph.D.)
School of Pharmacy and Health Sciences
International Medical University
126, Jalan 19/155B, Bukit Jalil
57000 Kuala Lumpur,
Malaysia
Tel : (+6) 0122085170
Fax : (+6) 0386567229
References